Author's response to reviews

Title: Reduced coronary flow reserve in Anderson-Fabry disease measured by transthoracic Doppler echocardiography

Authors:

Pawel Petkow Dimitrow (dimitrow@mp.pl)
Marek Krzanowski (mmkrzano@cyf-kr.edu.pl)
Anetta Undas (mmundas@cyf-kr.edu.pl)

Version: 2 Date: 21 April 2005

Author's response to reviews:

Dear Editor
We greatly appreciate valuable comments from reviewers. The changes in the text are marked by bold fonts.
Sincerely
Pawel Petkow Dimitrow